

# What about thematic information? An analysis of the multidimensional visualization of individual mobility

Aline Menin, Clément Quere, Jorge Wagner, Sonia Chardonnel, Paule-Annick Davoine, Wolfgang Stuerzlinger, Carla Maria Dal Sasso Freitas, Luciana Nedel, Marco Winckler

## ▶ To cite this version:

Aline Menin, Clément Quere, Jorge Wagner, Sonia Chardonnel, Paule-Annick Davoine, et al.. What about thematic information? An analysis of the multidimensional visualization of individual mobility. Visual Informatics, In press, 10.1016/j.visinf.2025.02.002 . hal-04944676

# HAL Id: hal-04944676 https://hal.science/hal-04944676v1

Submitted on 13 Feb 2025

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial - NoDerivatives 4.0 International License

# Graphical Abstract

## What about thematic information? An analysis of the multidimensional visualization of individual mobility

Aline Menin, Clément Quere, Jorge Wagner, Sonia Chardonnel, Paule-Annick Davoine, Wolfgang Stuerzlinger, Carla Maria Dal Sasso Freitas, Luciana Nedel, Marco Winckler

AUTHOR AND



# Highlights

## What about thematic information? An analysis of the multidimensional visualization of individual mobility

Aline Menin, Clément Quere, Jorge Wagner, Sonia Chardonnel, Paule-Annick Davoine, Wolfgang Stuerzlinger, Carla Maria Dal Sasso Freitas, Luciana Nedel, Marco Winckler

- Analysis of 38 visual analytics solutions designed to aid the exploration of individual mobility data across spatial, temporal, and thematic dimensions.
- An analysis of how visualization tools incorporate the theme dimension and which thematic properties are addressed.
- Analysis of how the subject is approached within GIS and VIS communities.



# What about thematic information? An analysis of the multidimensional visualization of individual mobility

Aline Menin<sup>a</sup>, Clément Quere<sup>a</sup>, Jorge Wagner<sup>b</sup>, Sonia Chardonnel<sup>c</sup>, Paule-Annick Davoine<sup>d</sup>, Wolfgang Stuerzlinger<sup>e</sup>, Carla Maria Dal Sasso Freitas<sup>b</sup>, Luciana Nedel<sup>b</sup>, Marco Winckler<sup>a</sup>

 <sup>a</sup> Université Côte d'Azur, CNRS, Inria, Av. Valrose, Nice, 06000, Alpes-Maritimes, France
<sup>b</sup> Federal University of Rio Grande do Sul, Av. Bento Gonçalves, 9500, Porto Alegre, RS, 91501-970, Brazil
<sup>c</sup> Univ. Grenoble Alpes, CNRS, Science Po Grenoble, PACTE, 14 bis, avenue Marie Reynoard, Grenoble, 38100, Auvergne-Rhône-Alpes, France <sup>d</sup> Univ. Grenoble Alpes, CNRS, Grenoble INP, LIG, 700 Av. Centrale, Saint-Martin-d'Hères, 38401, Auvergne-Rhône-Alpes, France
<sup>e</sup> Simon Fraser University, 8888 University Dr W, Burnaby, BC V5A 1S6, Canada

#### Abstract

This paper reviews the literature on the visualization of individual mobility data, with a focus on thematic integration. It emphasizes the importance of visualization in understanding mobility patterns within a population and how it helps mobility experts to address domain-specific questions. We analyze 38 papers published between 2010 and 2024 in GIS and VIS venues that describe visualizations of multidimensional data related to individual movements in urban environments, concentrating on individual mobility rather than traffic data. Our primary aim is to report advances in interactive visualization for individual mobility analysis, particularly regarding the representation of thematic information about people's motivations for mobility. Our findings indicate that the thematic dimension is only partially represented in the literature, despite its critical significance in transportation. This gap often stems from the challenge of identifying data sources that inherently provide this information, necessitating visualization designers and developers to navigate multiple, heterogeneous data sources. We identify the strengths and limitations of existing visualizations and suggest potential research directions for the field.

Keywords: individual mobility, information visualization, spatio-temporal

Preprint submitted to Visual Informatics

February 13, 2025

#### 1. Introduction

The study of human mobility has always been of great importance, whether to help understand the dispersion of humankind (e.g., migrations of *Homo sapiens* out of Africa, colonization of the New World) or to support decisionmaking processes that impact the lives of the individuals in a population (e.g., estimating migratory flows, traffic forecasting, urban planning, and epidemic modeling) [1]. The term "human mobility" refers to the movement of individuals and groups, but only individual movement allows for a detailed analysis of population displacement over time and space. By monitoring these movements, we can uncover personal motivations for mobility, enhancing our understanding of its impact on quality of life and spatial usage. Unlike earlier urbanization models that separated production and residential areas, leading to daily commutes to job-rich locations [2], modern mobility is shaped by a broader range of factors. Today, it is influenced not only by employment opportunities but also by leisure, tourism, and changing lifestyles, which vary significantly depending on individuals' age and geographic origins [3].

Historically, trip-based surveys have been used to assess the relative performance of transformation alternatives [4]. Activity-based surveys have also been a great source of information to unveil the links between trips and activities that explain the individuals' need for travel [5]. Thanks to recent technological advances, the wide adoption of the Global Positioning System (GPS), and the ubiquitous use of tracking technologies in personal devices, large-scale data collection for individual mobility has become easier. In the last decades, we have witnessed an exponential increase in the number, diversity, and size of datasets describing human mobility through multiple and diverse sources, such as household travel surveys (HTS), GPS-assisted surveys, or geo-referenced activity data. This multitude of data sources makes the design of effective visualizations challenging, due to the huge volume and multi-dimensional nature of the data.

Individual mobility data is inherently spatio-temporal (i.e., objects are defined over space and time) but often includes *thematic* properties such as demographics, transportation modes, and trip purposes that describe the individuals and their trips, respectively [6, 7, 8]. In this paper, we investigate how *thematic information* is represented in the (geo-)visualization literature,

and how it has been used in decision-making processes within urban policies (such as transportation options, accessibility, air quality control, public health, or well-being).

This work focuses on individual mobility, and it serves as a complement to previous surveys that have mainly focused on the traffic data [9], the use of vehicles and transportation systems [10, 11], or urban mobility more broadly [12, 13]. However, while the literature extensively discusses the visual exploration of mobility data across space and time, it lacks investigation of the thematic dimension that supports in-depth activity-based analyses of individual mobility data. This paper provides a comprehensive analysis of the thematic dimension of individual mobility data by addressing the following research questions:

- RQ1 How are the existing data sources leveraged to facilitate individual mobility analysis? We seek to identify the different data sources used to support the visual analysis of individual mobility, as well as the advantages and limitations of each one to support the answering of domain-specific questions.
- **RQ2** How do visualization systems incorporate the thematic dimension? We seek to identify to what extent and through which interaction and visualization strategies, current solutions integrate space, time, and the thematic dimensions.
- **RQ3** Are domain-specific tasks sufficiently supported by current information visualization systems? We seek to identify the different domain-specific tasks supported by existing visualization solutions. In particular, we focus on the relationship between thematic properties and the analysis of presence dynamics, travel flows, and individual trajectories (see Section 5).
- **RQ4 How are visualizations evaluated for usability?** We investigate how the surveyed papers evaluate the visualization solutions in terms of methods, experimental design, and the participant's sample and profile.
- **RQ5** How do GIS and VIS communities approach the visualization of individual mobility data? With research becoming increasingly interdisciplinary, we aim to identify the various contributions of the

VIS (Information Visualization) and GIS (Geographic Information Science) communities towards our topic and explore whether and how their approaches differ or overlap.

The remainder of this document is organized as follows. Section 2 describes how we approach the literature to answer our research questions. Section 3 presents an overview of data sources and their classification according to geographic, thematic, and temporal information. Section 4 presents an analysis of the visualization and interaction with individual mobility data. Domain-specific questions are addressed in section 5. Section 6 presents studies describing the uses and the usability of visualization techniques. Section 7 analyses the distribution of articles between VIS and GIS venues. Section 8 summarizes the survey's outcomes, while identifying directions for future research. Lastly, Section 9 presents limitations and conclusions.



2. Methodology

Figure 1: Overview of the survey's methodology.

To assemble a comprehensive review of existing studies, we followed a reference-driven and search-driven approach [14], as shown in Figure 1. We started from a core set of 35 state-of-the-art papers on the topic, encompassing publications from 2010 to 2018 [15], initially compiled by the authors through their previous works on the domain. From this core, we extracted a set of relevant search keywords: visualization, urban mobility, human mobility, individual mobility, space-time cube, trajectory visualization, mass mobility, and mobility data. We used those keywords as input to a two-round coarse search following a search-driven approach. In the first round, we used Google Scholar's search engine to look for papers published between 2010 and 2024 that included any combination of the defined keywords. The last search dates from October  $1^{st}$ , 2024, which yielded 1,211 papers. For each keyword combination, we recovered the 80 first papers, sorted by relevance. After reviewing the abstracts and full texts, we selected 27 papers. The second round consisted on a venue-focused search using both Google Scholar's advanced search engine and *Scopus*. For that, we tailored our search to target papers published in the identified journals and conferences of the previous phase (see the list of venues in Table 7). This search resulted in 1,439 papers. The review of abstracts and full texts of the new papers resulted in the selection of 11 additional papers, comprising a total of 38 papers. The following exclusion criteria were applied to the selected papers:

- Papers not written in English.
- Duplicate papers.
- Visualizations not intended for end users. We exclude papers that do not mention the use of visualization by users, which often feature research questions unrelated to information visualization.
- Secondary studies. We focused on the review of primary studies and thus did not include surveys.
- Absence of figures. We only consider papers that demonstrate visualizations through illustrations.
- Papers that use simulated data. We only include papers that use real-world data. Artificially generated datasets were excluded as they often hide the inherent heterogeneity, sparsity, and subjectivity of real-world data.

• Non-individual mobility data. We only include papers that use data sources describing human mobility data either at the individual level (e.g., household travel surveys) or allowing it to be derived (e.g., smart card data or telecommunication).

After screening the abstracts and full texts, we selected 38 papers published between 2010 and 2024, with a notable peak in publications occurring between 2015 and 2018 (Figure 2). This relatively small number of publications underscores the limited attention given to the visualization of individual mobility data in the academic literature, irrespective of the publication venue. Hereafter, we present the selected papers, organized according to five aspects: input data, domain-specific questions, interactive visualization, evaluation aspects, and publication venue.



Figure 2: Distribution of the surveyed papers by publication year.

#### 3. Input Data

To answer **RQ1**, we identified the different data sources used in the papers, and classified them according to the collection method, structured by the dimensions of individual mobility data [8]:

- **Space**: the geographic information, for which we identify the (i) spatial positions (i.e., area, origin-destination (O/D), or trajectory), (ii) granularity (i.e., division-, linear-, or coordinate-based referencing [16]), and (iii) dimension (i.e., 1D, 2D, 3D).
- **Time**: It can be described as (i) an event (e.g., a planned public or social occasion, catastrophes, etc.), (ii) a duration (i.e., a period without precise beginning or end), (iii) an interval (i.e., a period with known beginning and end), or (iv) a timestamp.
- **Theme**: It describes the thematic properties that characterize individuals and trips. The theme includes (i) demographic information, (ii) transportation modes, and (iii) trip purposes, for which we distinguish granularity (i.e., 0, 1, or N) and associated values.

Table 1 presents the different types of data sources identified within the surveyed papers, along with the data dimensions they describe. They are defined as follows:

- Surveys provide datasets describing single trips through space (i.e., origin and destination locations, and possibly travel route), time (i.e., start/end), and thematic information (e.g., trip purpose, transportation mode, and traveler information) [4]. Household Travel Surveys are the most common type of survey, capturing household members' travel practices through trip purposes and transportation modes. Alternatively, Qian et al. [17] used the Longitudinal Employer-Household Dynamics Origin Destination dataset, which describes (O/D) commuting trips within an urban environment.
- Mobile phone data (MPD) are collected via internet connections, cell towers, or applications running on mobile phones [18]. Call Detail Records (CDRs) are frequently used. Records generated by telephone exchanges provide details of phone calls, such as start/end time, duration, caller identifier, and O/D towers [19]. Telco data (exchange records between mobile phones and cell towers) can be used to analyze when users make calls, send messages, or access the internet. Senaratne et al. [20] extracted mobility patterns from Global System for Mobile Communications records, which provide data on network traffic, antenna operator, location, cell identifier, signal strength, and phone activity status.

- Public Transportation System (PTS) Data describes mobility via PTS usage. A popular data source in this category is *Smart card data* (*SCD*) from validation cards used in subways and buses, which provide information on where and when a passenger tapped in/out, i.e., the O/D locations describing each individual trip. Similarly, *Bike-sharing data* offer details on when and where bikes were rented and returned, along with information about the renter, such as age.
- Social media data is provided by applications such as Twitter or Sina Weibo, which allow one to announce their arrival at a hotel, airport, hospital, and so on. The data is often semantically enriched with users' impressions on the place or an event linked to that location [21].
- Traffic sensors data (TSD) is collected through traffic sensors strategically placed over an urban area to record pedestrians, bicycles, cars, and other vehicles passing a particular street. These sensors can detect a vehicle moving and estimate the speed, time, and direction of the vehicle.
- **GPS** data (collected via a GPS receiver) provides latitude and longitude information allowing one to geo-locate a person on the earth's surface. In this category, we identified Taxi and personal vehicle data.
- Point of Interest (POI) data provide thematic information describing spatial locations that may be of interest to individuals, businesses, or applications. It is typically used to derive the function of visited places and, thus, estimate trip purposes.

Data Source	Space	Theme	Time
Survey	O/D	Travel Mode, Trip Purpose, Demographics	Start/End (HH:MM)
Mobile Phone Records	Area	None	Timestamp
Public Transportation System	Point	Travel Mode	Timestamp
Social Media	Point	None	Timestamp
Traffic Sensor Data	Point	None	Timestamp
GPS	Trajectory	None	Timestamp
Point of Interest	Point	Locations' description	None

 $\boldsymbol{\lambda}$ 

Table 1: Overview of data sources classified according to geographic, thematic and temporal information they provide.



Figure 3: Distribution of data capture methods according to the type of information described (i.e., mobility or theme), the type of data source, and the spatial and thematic information provided.

Every data source considered in this survey contains individual movements over space and time. However, the level of granularity of information depends on the method used to capture data. The temporal information is always represented through timestamps, while spatial information exhibits variability in terms of location (area, individual point, or origin/destination points), granularity (ranging from division-based, coordinate-based, to linear representations), and dimensionality (1D or 2D). The diagram in Figure 3 shows the distribution of 18 unique capture methods identified according to the type of information they describe (i.e., movement or thematic information), the types of geographic and thematic information provided, and the type of data source.

It is worth noting that the thematic information needed to explain people's motivations for mobility is often absent from the data sources. Surveylike data sources are nearly unique in gathering thematic and mobility information. Thus, combining data from diverse sources is a growing practice, observed in 14 of the 38 surveyed papers. In particular, the goal is often to (i) complement the data with thematic information such as trip purpose and transportation mode, a practice observed in previous work [22, 23, 17, 24] or (ii) increase data accuracy, such as by combining O/D segments from SCD with bus line and station data to derive the actual trajectory traveled by each person [25]. Thematic data from real estate data has been used to support filtering operations, such as to focus the analysis on neighborhoods characterized by high and cheap housing prices respectively [17].

Although most of the papers (n=24) do not explicitly mention the volume of data, the information we could gather indirectly still gives a glimpse into the magnitude of data processed for the visualizations. The reported values span from 119 to a staggering 272 million trips (M = 24 million trips). Additionally, 27 papers in our survey provide information about the number of individuals whose movements are documented within the dataset. These figures fluctuate between 180 and 88.9 million individuals (M =3.2 million people).

Regarding the type of data, the majority of papers rely on O/D segments (constituting 44.7% of the papers), followed by trajectories (39.4% of papers), and events (featured in 13.1% of papers). Over half of the papers explore the spatial information at the level of geographic coordinates, usually to represent individual trajectories or spatial distribution of people, while 8% aggregate the data at the level of administrative partitions, and the remaining ones adopt custom spatial division depending on their analysis (e.g., Voronoi partitions [26, 27], hierarchical grids [23]), or the data source (e.g., household travel surveys often describe the space through custom pooling areas based on the population to ensure statistical significance [28]). The time dimension is frequently treated at the level of intervals of one to three hours (60% of papers), or at the level of timestamps (36% of papers), typically when representing individual trajectories over time. Other aggregations are used according to the analysis, such as dividing the day into periods that describe the morning peak, the afternoon, and the night valley [22]. While the majority of data sources lack intrinsic information regarding the thematic attributes of trips and individuals, it is noteworthy that 26 out of the 38 examined papers do incorporate such information. Often, these thematic attributes are derived from external data sources to enrich the analysis and understanding of the mobility data.

Paper	Trip	Travel	Demogra-	Travel	Travel/Stay	Travel
1 aper	Purpose	Modes	phics	Distance	Duration	Speed
[29]						
[30]						
[24]						
[31]						
[32]						
[33]						
[34]						
[35]						
[23]						
[25]						
[36]						
[37]						
[38]						
[39]						7
[40]						
[41]						
[22]						
[42]						
[43]						
[44]						
[45]						
[21]						
[46]			7			
[47]						
[20]						
[18]						

Table 2: Classification of the 26 papers that include the theme dimension based on the properties they represent.

## 4. Visual Mapping and Interaction

Individual mobility data is intrinsically multidimensional. Thus, diverse visualization techniques can be used alone or in combination to support analytical tasks, such as identifying the distribution, comparisons, and relationships among the different dimensions of the data. In the surveyed papers the number of visualization techniques vary from one to seven, with a median of 4 techniques per visualization system. Two papers employ a single technique for visualization (i.e., the space-time cube and isochronous map). To address **RQ2**, we classified the visualization techniques according to the supported functions and data dimensions, with particular focus on how the theme dimension is addressed.

We did not consider a timeline as a visualization technique, as it was mostly used to filter the data. We identified a total of 25 distinct visualization techniques. Table 3 presents each technique and the user tasks they support. In general, the majority of the techniques support fundamental analytical tasks, such as pattern recognition (36 papers), distribution analysis (34 papers), location identification (33 papers), comparison (33 papers), exploring relationships (28 papers), representation of data over time (26 papers), and movement or flow visualization (24 papers). Notice that different visualization techniques are used to address similar analytical tasks.

Space and time form the foundation for nearly every visualization designed to communicate individual mobility data. Representation-wise, there is little novelty: space is often represented via geographic maps, such as connection maps, flow maps, and choropleth maps, while time is represented either via time-juxtaposing or animation. Four studies explored the usage of abstract visualizations through Voronoi diagrams [27, 26] and treemaps [48]. Only one study does not provide a spatial representation of the data, as they focus on exploring individual activity patterns over time [36].

Figure 4 depicts the distribution of papers over the years according to the type of data used to represent people's movements, the data source, and the thematic representation, in terms of data and visual variables. Table 2 presents the diversity of thematic attributes considered by those papers. The most commonly represented property is trip purpose (16 papers), followed by travel mode (8 papers), individuals' demographics and travel distance (5 papers), travel or stay duration (3 papers), and travel speed (2 papers). Five papers (i.e., [29, 30, 32, 24, 31]) visually depict multiple thematic properties, whereas the others typically focus on a single thematic property, using the rest as filtering criteria. Particularly, demographics information are often used as a means of data filtering to support domain-specific tasks, such as understanding the mobility patterns of particular ethnic groups [45]. When visually integrated, thematic information is primarily depicted through visual cues like color, texture, or icons, integrated into visualization techniques that already represent space, time, or both. Hereafter, we summarize the visualization techniques employed by the surveyed papers, emphasizing how they integrate the theme dimension.

Thematic Maps. Thematic maps often convey thematic information that characterizes the usage of different regions within the studied territory. Choropleth and bubble maps have been used to represent mobility metrics such as the spatial distribution of stays per activity [30, 29], thus revealing the reason people occupy the different regions of the territory. Similarly, density maps represent people entropy (i.e., the magnitude of resources a person is con-

Dim.	Technique	Analytical Task	Papers
	flow map	view the distribution of OD flows	[22, 49, 37, 41, 18, 50,
Space	now map		46]
	heatmap	population	[22, 20, 50, 38, 40]
	space-time cube	view individual trajectories or OD relationships	[32, 45, 27, 15, 51, 37, 22, 20, 24]
			$\begin{bmatrix} 33, 20, 24 \end{bmatrix}$ $\begin{bmatrix} 23, 25, 45, 21, 27, 52 \end{bmatrix}$
	connection map	view individual trajectories or OD relationships	53, 26, 37, 33, 41, 31, 50,
	density map	view people or transportation mode distribution	$\begin{bmatrix} 12, & 10, & 11 \\ [25, & 21, & 27, & 48, & 54, & 24, \\ 41, & 31 \end{bmatrix}$
	dot map	view people distribution	$\begin{bmatrix} 45, 52, 53, 37, 41, 50, \\ 34 \end{bmatrix}$
	spatial 3D bar	view the variance of phone calls or areas of activity	[27, 37]
	choropleth map	view people distribution, density or fluctuation	[27, 15, 17, 54, 30, 44]
	bubble map	view variance of people presence	15, 41
	isochrone map	view OD nows view reachable areas	
	contour-based	view starting locations of incoming trips	[48]
	treemap sankey diagram	view OD flows	[42]
	network diagram	view OD relationships	[42]
T:	space-time cube	view individual trajectories	[32, 45, 27, 15, 51, 37,
Time	glyph	view the weekly distribution of daily stays	$\begin{bmatrix} 33, 20, 24 \\ [23, 40, 38] \end{bmatrix}$
	8-7 F	view the distribution of trips, mobility descriptors (e.g.	25, 21, 37, 49, 33, 53,
	nistogram	speed, distance) or infection risk	24, 41, 18, 44, 46, 47]
	chart	view the distribution of people or trips	[25, 21, 15, 33, 18, 50]
	stacked-bar chart	view the distribution of activities or transportation	[23, 36, 15]
	activity-time cube	view the distribution of activities	[36]
	parallel sets	view mobility transition patterns	[53]
	stacked-area chart	tion modes over a day	[15, 26, 33, 31, 38, 40]
	line chart	view the distribution of people or mobility descriptors (e.g. speed distance)	$\begin{bmatrix} 17, 52, 20, 33, 30, 24, \\ 34, 46 \end{bmatrix}$
	contour-based	view the distribution of visitors and visiting locations	[48]
	treemap	view the distribution of visitors and visiting locations	[==0]
	radial chart	view the distribution of visitors and visiting locations	[40]
	sankey diagram	view the distribution of OD flows, visitors over a day	42
	alluvial diagram	view the distribution of OD flows, visitors over a day	[40]
Theme	heatmap	tance (color)	[22, 21, 39]
	scatterplot	represent distance (x-axis) and transportation mode (color)	[22]
	space-time cube	represent socioeconomic aspects or trip purpose (color)	[32, 45, 15, 33, 20, 24]
	glyph	represent trip purpose and transportation modes	[23, 44, 46, 40]
	multi-ring donut	represent trip purpose, transportation mode or socio-	
	chart	demographic aspects (color)	[20, 21, 10, 35, 44]
	stacked-bar chart	demographic aspects (color)	[36, 15, 33, 31]
	activity-time cube	represent trip purpose (color)	[36]
	parallel coordinate	represent socio-demographic aspects (axes) and ethnic	[45, 31]
	dot map	represent ethnic groups or trip purpose. POI (color)	[45, 37, 38, 35]
	connection map	represent ethnic groups (color)	45
	histogram spatial 3D bar	represent POIs (x-axis) and transportation modes	[54, 24, 44]
	chart	represent activities (color)	[27, 37]
	bubble map	represent activities (color)	15
	line chart	represent activity (color)	[33, 34]
	semantic chart	represent activities and transportation modes	
	pie chart	represent trip distribution over trip purpose represent activities and transportation modes	38

Table 3: Overview of visualization techniques and supported user tasks and data dimensions by each paper.

nected to and the commutation frequency in daily jobs) and segment entropy (i.e., the diversity and fluidity of spatial locations based on the visits' length) metrics [54]. Gortana et al. [43] superimposed multiple isochrone curves over a map to display reachable distances for a particular spatial location accord-



Figure 4: Distribution of papers according to the data type used to describe movements, the data source, the thematic information with the visual variables used to represent it, and the publication year.

ing to different transportation modes, travel time, and hours of the day. Yu et al. [37] represented spatial locations (i.e., home and workplaces) using a dot map, where a given spatial location (shown as a dot) can be selected to reveal the incoming and outgoing flows.

Flow-based representations. While line thickness is conventionally used to represent flow magnitude in flow maps and chord diagrams, it has also been used to represent trip distance [46] (Fig. 5A). In these techniques, the color has been used to encode thematic properties such as transportation modes [22, 29], flow direction [49, 55] (Fig. 5C), trip purpose [37, 29], or trip frequency [46]. Pérez-Messina et al. [22] used an O/D matrix to represent travel flows between spatial locations, which cells are color-coded based on the trip's transportation mode (e.g., private, public, or active). Zeng et al. [42] employ a three-component Sankey diagram to simultaneously illustrate the magnitude of flows between O/D locations, with the corresponding metro lines represented by color (Fig. 5B).



Figure 5: Thematic integration in flow- and time-based representations. (A) Sankey diagram [42]. (B) Flow map [46]. (C) Chronogram [23]. (D) Chord diagram [55]. (E) Multiring donut chart [33]. (F) Contour-based Treemap View [48]. (G) Multi-ring stacked-area chart [44]. (H) Semantic graph [31]. (I) Parallel coordinates plot [56].

Linear time-based representations. Line charts and histograms are used to represent the distribution of stays [17], trip count [52, 33], segregation index [30], and mobile network usage amounts [20]. In this context, color has been used to encode thematic information, such as the purpose of trips [33, 49] and trip direction [37, 33]. Stacked-area charts are often used to represent the distribution of trips over time per trip purpose [29, 30, 33], transporta-



Figure 6: Glyph-based representations commonly placed on top of geographical maps: (A) distribution of an individual's activities over the day and week [23], (B) the proportion of arrivals, departures, and stays [49], (C) the distribution of arrivals, departures, and stays via taxi and metro [44], and (D) the distribution of arrivals, departures, and activities [38].

tion mode [40, 29, 30] and travel speed [26]. Similarly, chronogram techniques are used to represent the sequence of activities or transportation modes that comprise the trajectory of a person during the day [29, 36, 23] (Fig. 5D).

Treemaps. Traditionally used for representing hierarchical data, treemaps have been employed in innovative ways to illustrate individual mobility data. Wu et al. [48] present a *Contour-based Treemap View* that combines treemaps and a radial chart to visualize human co-occurrence (Fig. 5E), displaying the spatial origins of visitors at a selected location. The radial chart is divided into seven sectors representing 2-hour intervals from 6 am to 8 pm, and a sector for the nighttime period from 8 pm to 6 am. Each sector corresponds to the arrival time of visitors in a region indicated as a rectangle inside the sector. The size of the region encodes the number of visitors, while the color indicates the distance between the origin and the destination of visitors.

Cyclic time-based representations. Inspired by clock designs, multi-ring donut charts have been used to visualize the distribution of stays or trips over timebased on factors such as trip purpose or transportation mode [29, 23], as well as socio-demographic characteristics like gender, age, and occupation [33], either for specific spatial locations or across an entire area (Fig. 5F). This method involves multiple concentric donut charts, each divided into sectors representing time intervals. These sectors are either color-coded to indicate a particular thematic variable or further subdivided to display the distribution of stays or trips for each variable within that period. In a variation of this approach, Wu et al. [44] use stacked-area charts in place of color-coded sectors to depict the distribution of trips in- and outbound, via metro or taxi, alongside the fluctuation of stays throughout the day (Fig. 5G).

Theme-specific representations. Parallel coordinates plots are effective for representing multiple attributes simultaneously and have been utilized to illustrate the variation of trips across POIs [56] (Fig. 5H), trip aspects such as distance, number of people, and time periods [48], and demographic characteristics [45]. In the latter, factors like age, language, education level, and income are plotted as vertical axes, with color-coded lines connecting them to represent different ethnic groups [45] (fig ref). Scatterplots have also been employed to depict the influence of geographic distance and transportation mode on travel flows [22], where the y-axis indicates the number of individual trips and the x-axis represents varying travel distances. The intersecting values are illustrated by circles, with color and size encoding transportation mode and trip count, respectively. Sankey diagrams have been used to represent the variation of trips over trip purpose [39], Additionally, Huang et al.[31] introduced a novel technique called the *semantic graph*, which represents individual trajectories as polylines within a topic hexagon (Fig. 5I). where the vertices correspond to points of interest (POIs). This design enables users to identify (i) trajectories associated with specific purposes and (ii) the purposes that drive varying activity levels.



Figure 7: 3D-based representations. (A) Space-time cube (STC) for GPS-type trajectories with color encoding inter-sample time interval [20]. (B) STC for O/D trajectories with color encoding activities [15]. (C) Activity-time cube representing activity sequences over time [36].

*Glyph-based representations*. As noted by Borgo et al. [57], due to their ability to convey multiple attributes within a compact visual space, promoting enhanced cognitive engagement during the visualization process, glyphs are popular for representing individual mobility data. A common practice observed in the surveyed papers is to position glyphs on a map based on geographic coordinates, thereby characterizing a location by the temporal distribution or aggregation of attributes such as trip purpose, transportation mode, or trip direction. For instance, Chen et al. [23] proposed a glyph in the form of a multi-ring donut chart to represent the temporal distribution of activities of an individual at a particular location (Fig. 6A). Often, glyphs are used to represent the repartition of trips in and out of a particular spatial location [46, 49, 44, 40, 38]. For instance, Silva et al. [49] proposed a donut-shaped glyph with a filled center to characterize spatial locations as a local activity, departure, arrival, or pass-by point according to the trip direction and destination (Fig. 6B). A stacked elliptical glyph design [44] was proposed to summarize and compare activities in different spatial locations according to number of stays and trips per transportation mode (Fig. 6C). Cyclic stacked-area charts are employed to visualize the temporal distribution of activities [40] as well as the arrivals and departures of people [38]. Zeng et al. [38]'s glyph positions the stacked-area chart along the circumference of a pie chart, representing both the arrivals and departures of people and the distribution of activities at a given location (Fig. 6D).

3D-based representations. Due to the multidimensionality of individual mobility data, 3D representations are becoming popular. A frequent example is the space-time cube (STC) [58], a technique that represents geographical space on the cube's base and time along its height. In nine of the surveyed papers, the STC has been used to simultaneously represent space, time, and thematic properties through symbols and colors. For instance, Kveladze et al. [45] used the technique to represent individual trajectories across socioeconomic aspects, such as ethnicity and education level. Senarate et al. [20] used the STC to inspect data uncertainty by coloring the individual trajectories according to the inter-sample time interval (i.e., the time interval between two data records) (Fig. 7A). The STC by Yu et al. [37] supported the exploration of mobility patterns by representing individual trajectories or clusters, where color encoded the trip direction (i.e., going home or to work). In general, the STC was used to represent individual activity patterns across space and time, where color encoded the different activities or trip purposes [29, 33, 32] (Fig. 7B). Alternatively, the *activity-time cube*, proposed by Vrotsou et al. [36], encoded the temporal distribution of activities per individual through a three-dimensional space, where the x-axis holds the individuals, the y-axis is the time axis and the z-axis can display activities,

places, or companionship (Fig. 7C).

User Task	Mobility elements / indicators	Papers
	Data and view parameters	$\begin{matrix} [45, 21, 15, 17, 52, 43, 49, 48, 54, 51, \\ 26, 24, 41, 31, 18, 44, 40, 56, 46, 35, \\ 39, 47 \end{matrix}$
Select	Time periods	$\begin{bmatrix} 23, 32, 45, 21, 29, 17, 52, 43, 49, 48, \\ 54, 53, 26, 30, 20, 24, 42, 40, 47 \end{bmatrix}$
	Spatial locations	[29, 52, 48, 54, 26, 37, 20, 24, 31, 50, 38, 40, 56, 47]
	Individuals trajectories or trips	[22, 25, 36, 27, 29, 52, 49, 53, 31, 50, 42, 47]
	Individuals trajectories or trips	$\begin{matrix} [23, \ 32, \ 45, \ 21, \ 29, \ 17, \ 43, \ 49, \ 51, \ 53, \\ 31, \ 50, \ 42, \ 39, \ 47 \end{matrix}$
Filter	Socio-demographic aspects	[48, 26, 30, 29, 23, 25, 32, 30, 34, 44, 40, 56, 46]
	Spatial locations	[32, 29, 45, 21, 54, 53, 26, 24, 38]
	Time periods	[54, 29, 24, 41, 31, 18, 55, 46]
	Mobility indicators	[29, 54]
	Purposes	
	Mobility flows or trajectories	[22, 23, 27, 52, 49, 26, 37, 33, 24, 31, 18, 50, 55, 56]
Change over Time	Mobility indicators	[29, 48, 54]
	Travel time	[43]
	Socio-demographic aspects	[30]
Pan	Map view	[36, 32, 45, 21, 29, 52, 43, 49, 51, 24,
1 0011	map view	44, 40, 56, 46, 47
		[23, 36, 32, 45, 21, 27, 29, 17, 52, 43,
Zoom	Spatial locations	48, 51, 53, 20, 24, 31, 18, 50, 44, 40,
		56, 46, 47
	Get more or less details	[49, 41, 55]
Rotate	Camera	[36, 32, 45, 29, 51]
Reorder	Mobility patterns	[21, 53, 42, 34, 40]
	Sort activities	[ [38, 35]
	Spatial locations	
Annotation	Identify activities	[ [38, 35]
	Spatial locations	[34, 31]

Table 4: Classification of papers based on user tasks and the mobility elements or indicators they address.

Interaction. Table 4 provides an overview of the user actions supported by the different visualization tools as well as the data items (e.g., spatial locations, time units, etc.) or mobility indicators (e.g., activity patterns, presence density, etc.) that can be manipulated through user actions. Overall, the visualization tools support familiar analytical tasks-including selection, filtering, zooming, change over time, panning and rotation, reordering, and annotation-which are widely recognized in visualization [59]. Most papers rely heavily on the mouse and keyboard for physical interaction, even in 3D visualizations. However, one study [29] explores spatial interaction using a tablet's tilting motion to control time-based changes. In this approach, tilting angles are mapped to time units on a timeline, which then animates a thematic map.

#### 5. Domain-specific questions

To answer **RQ3**, we categorize the papers based on three common questions frequently raised by mobility experts, including geographers and urban planners: (i) What are the commuting and travel patterns of individuals within urban areas and how do these patterns influence the configuration of the urban landscape?, (ii) How do the socioeconomic attributes of individuals and the activities they engage in influence land usage within urban regions?, and (iii) How does the geographic context of a given area influence the scheduling of individuals' activities and journeys throughout the day? [15]. We further classify the papers according to the addressed mobility issue and the different mobility indicators derived from the data. Using the pyramid framework introduced by Mennis et al. [8] as a foundation, we classified the queries into four distinct categories that define the connections between the different aspects of the data needed to support users in addressing domain-specific questions. These queries are defined as follows:

- when + where + theme → what: describe a particular (set of) object(s) based on a particular (set of) thematic attribute(s) existing at a given (set of) location(s) at a given (set of) time(s);
- when + what + theme → where: describe a particular (set of) location(s) occupied by a given (set of) object(s) based on a particular (set of) thematic attribute(s) at a given (set of) time(s);
- where + what + theme  $\rightarrow$  when: describe a (set of) time(s) that a particular (set of) object(s) occupied a given (set of) location(s) based on a particular (set of) attribute(s); and
- when + where + what → theme: give the (set of) thematic attribute(s) describing the (set of) object(s) existing at a given (set of) location(s) at a given (set of) time(s).

Table 5 presents the surveyed papers classified by domain-specific task, according to the above taxonomy. The papers mostly focused on exploring human behavior within an urban environment in terms of when people

Type	Task	Papers
01	Identify <b>who</b> is performing an activity at a given place and time	[29, 37, 32, 23, 38]
	Identify demographic groups visiting a location at a specific time	[25, 45, 30]
	Compare <b>trajectories</b> over space, time and demographic groups	[29, 45]
when $+$ where $+$ theme	Identify <b>trips</b> using a given travel mode at a given place and time	[22, 29]
$\rightarrow$ what	Compare individual <b>trajectories</b> per trip purpose	[37, 23]
/ What	Identify <b>trips</b> with a given purpose at a given place and time	[20]
	Identify events at given location and time	[56]
	Identify the <b>individuals</b> visiting a specific location at a given time	[29, 17, 30, 37, 32, 27, 53,
	identify the <b>individuals</b> visiting a specific focation at a given time	55, 42, 34, 40, 46]
when t where t wheet	Identify similar <b>trajectories</b> at a given location and time	$\begin{matrix} [49,21,52,27,20,45,51,\\ 24,37,39 \end{matrix} ]$
when $+$ where $\rightarrow$ what	Identify the number of <b>trips</b> at a specific location and time	$[33, 26, 30, 50, 31, 29, 18, \\20, 37]$
	Identify the number of <b>individuals</b> sharing the same O/D locations	[49, 29]
	Count the <b>trips</b> to or from a specific location at a given time.	[48, 49]
when $+$ theme $\rightarrow$ what	Count the <b>individuals</b> engaged in a given activity and time.	[23, 29, 36, 31, 30]
where $+$ theme $\rightarrow$ what	Count the <b>individuals</b> engaged in a given activity and location.	[23, 29, 30]
	Identify where people engage in a given activity and time	[29, 30, 37, 32, 34]
when $+$ what $+$ theme	Compare locations over time based on land use	[33, 29, 30, 44]
ightarrow where	Identify reachable <b>locations</b> by a given travel mode in a given time	[43, 22, 52]
	Identify <b>locations</b> visited by various demographic groups	[30, 45]
	Explore visitation patterns of spatial <b>locations</b> over time	$[49, 41, 18, 21, 27, 37, 31, \\32, 54]$
1 1 1 1 1 1	Explore movement patterns on <b>roads</b> (locations) over time.	[24, 18, 21, 26, 55, 46, 35, 47]
when $+$ what $\rightarrow$ where	Identify the locations where other locations co-occur	[48, 49, 50]
	Identify spatial <b>locations</b> where people co-occur	[48, 49, 40, 47]
what + theme $\rightarrow$ where	Identify working and residential locations	[18, 37, 56]
when $+$ theme $\rightarrow$ where	Characterize <b>locations</b> by travel modes at a given time	[22, 29]
where $+$ what $+$ theme	Identify when locations are visited for various purposes	[30, 29, 37, 32, 42]
ightarrow when	Identify when given demographic groups visit given locations.	[30, 45]
where   what \ when	Identify when given locations are visited	[29, 27, 21, 37, 32, 17, 48, 40]
where $+$ what $\rightarrow$ when	Identify when people co-occur in given locations	[48, 23, 34]
theme + what $\rightarrow$ when	Identify when individuals engage in given activities	[32, 29, 36, 23, 31, 33]
where $\perp$ theme $\rightarrow$ when	Identify when different demographic groups stayed at given locations.	[33, 29, 30]
where + theme -> when	Identify when people visit places for a given purpose or travel mode	[29]
	Identify and compare individual activity patterns	[29, 23, 37, 32]
	Compare <b>activity distribution</b> over space and time	[29, 30, 33, 38]
when + where + what $\rightarrow$	Identify the main <b>travel mode</b> to reach a location at a specific time.	[43, 29, 22]
theme	Compare <b>travel mode</b> distribution over space and time	[29, 44]
	Identify the <b>trip purposes</b> of trajectories	[31, 29]
	Compare the <b>demographic</b> share at a location over time.	[33]
	Identify the distribution of activities over time	
when + what $\rightarrow$ <b>theme</b>	Compare activity patterns over time	[29, 31]
	Identity shared activities at a given time	[36, 29]
	Compare <b>modal share</b> of trips for a given distance and time	
where $+$ what $\rightarrow$ <b>theme</b>	Identify the <b>activities</b> individuals engage in at given locations.	[33, 29, 30, 35]
	Identify the <b>demographic snare</b> of people at a given location	
when $+$ where $\rightarrow$ <b>theme</b>	Identify activities performed in a given location and time	[33, 29, 30, 46]
	Identify the <b>travel modes</b> to reach a given location at a given time.	[29, 30]

Table 5: Classification of papers according to domain-specific tasks.

visit different spatial locations or where people are at a given time. A few papers focus on understanding the commuting behavior of the population, which supports the identification of residential and working locations [37, 18]. Similarly, Le Roux and Vallée [30] supported a deeper analysis through the exploration of a segregation index revealing the temporal evolution of land use within the urban environment. According to the domain-specific questions, we classified the papers according to three distinct objects of interest: (i) presence dynamics, (ii) travel flows, and (iii) individual trajectories, respectively. Twenty-five publications represent presence dynamics through the variation of people present at different spatial locations and time intervals while eighteen support the visualization of travel flows and trips and sixteen represent individual trajectories of people or activity sequences.

Although every task in the taxonomy is supported at some level by the set of surveyed papers, a domain expert would have to perform an ad-hoc analysis by combining the visualization solutions provided by several different papers to be able to answer all three domain-specific questions. Only three visualization solutions [27, 37, 29] enabled the analysis of individual mobility data through the exploration of all three objects of interest. Both Gao et al. [27], and Yu et al. [37] employed 3D visualization techniques to simultaneously represent spatial and temporal information. However, these solutions have limitations when examining thematic data, which is restricted to describing the trip as going home or to work. Menin et al. [29] had a deeper focus on the spatiotemporal evolution of individual mobility across diverse thematic aspects. For instance, they covered trip purpose over eight categories (i.e., home, work, leisure, education, shopping, business, personal business, and escort trips, and travel modes over four categories (i.e., biking, driving, PTS, and walking), while providing a clustered exploration of individual trajectories based on daily activity sequences.

#### 6. Usability Evaluation

To address **RQ4** and understand whether and how user-based evaluations are employed to assess the usability and effectiveness of visualizations in the domain, we classified the papers according to their experimental design, the evaluated features (e.g., effectiveness, usability, satisfaction, etc.), the sample size, and whether the participants meet the profile of the target users of the application, i.e., mobility experts, including researchers in geography or transportation and urban planners. Table 6 presents the classification of papers according to these criteria.

The 38 papers analyzed in detail in this survey consistently provided some sort of validation of the proposed visual analytics system, mostly through case studies, where the visualization approach was applied to a real dataset or the usefulness of the approach was demonstrated through use case scenarios. For instance, Chen et al. [21] applied their approach to datasets describing the movements of tourists within two different cities, showing step-by-step how their system supports the discovery of knowledge such as typical tourist routes and popular destinations.

Twenty-two papers (about 60%) conducted user-based experiments to evaluate of their systems. Seventeen evaluations were conducted through a *non-experimental* design (i.e., without comparison [60]). Among these, fifteen studies interviewed a small group of experts (from 1 to 14 individuals). These experts were then presented with the system's interactive components, visual encoding, and a series of case studies and asked to provide feedback on the visual encoding and the suitability of the visualization for exploring the dataset. Three studies used a quasi-experimental design, where the visualization solution was compared across non-randomly assigned groups.

Experimental design	Sample size	Profile	Papers
Emonimental	7	expert	[45]
Experimentai	7-20	non-expert	[36, 45, 29]
	3-7	expert	[29, 40]
Quasi-experimental	24	quasi-expert	[29]
	10-20	non-expert	[29, 49, 40]
	1-11	expert	[22, 23, 48, 53, 26, 41, 31, 18,
	1-11	expert	38, 42, 44, 56, 47]
Non our originated	unknown	expert	[52, 50]
Non-experimental	4-14	non-expert	[24, 35]

Table 6: Classification of papers according to experimental design, sample size, and participants profile.

Three studies conducted formal experiments involving a larger sample of participants (from 7 to 20 individuals), Vrotsou et al. [36] investigated the efficiency of 2D (stacked-bar chart) and 3D (activity-time cube) representations of individual activity sequences, showing that search time and error rate increases proportionally to the number of activity sequences displayed in either technique. Kveladze et al. [45] investigated the performance of two distinct groups of users (i.e., expert and non-expert relative to the application domain) while exploring individual mobility data using a system comprising a 2D base map of the study area, a parallel coordinate plot, and four views on the 3D STC, each representing the trajectories of different ethnic groups of individuals. They observed that domain experts performed the tasks using logical and strategical reasoning while the non-experts did not follow the logical sequence of task execution. Probably due to more experience with the type of data and the different representations, the expert users also used all the available visualizations while the non-experts focused mainly on the STC, which the authors suggested was due to their enthusiasm toward the 3D visualization. Finally, Menin et al. [29] investigated the usability and effectiveness of a movement-based animation technique compared to the traditional one, and their results showed that both are equally effective for simple identification and comparison tasks.

Almost two decades ago, Ellis and Dix [61] highlighted the limited presence of user-based evaluations in the field. In their survey of over 65 papers, they found that only 12 of them included evaluations. In the context of individual mobility, we observed more promising numbers, although the number of formal experiments remains limited. The challenges associated with recruiting real users are evident in the limited group sizes of domain experts and the reliance on non-expert participants in experiments that require larger sample sizes.

#### 7. Publication Venues

The problem of visualizing individual mobility data can be addressed from multiple perspectives, which can vary across research communities. We hypothesize that the VIS community addresses the visual representation of multiple data dimensions while dealing with human cognitive and perceptual limitations, while the GIS community mostly addresses the challenge of understanding the dynamics of people and their impact on the urban environment, where visualizations are used to support the study, instead of being the main goal. Thus, to address **RQ5**, we classified the surveyed papers based on the type of venue where they were published, i.e., VIS or GIS.

Table 7 presents the papers according to the publication venue. The majority of papers were published in VIS venues (n=22, such as IEEE TVCG, PacificVis, IEEE VIS), or other subareas of Computer Science (n=3). Ten papers were published in GIS venues, such as the Journal of Transport Geography and IEEE Transactions on Intelligent Transportation Systems, and three papers were published in interdisciplinary venues (i.e., the International Conference on Cross-Cultural Decision Making and the International Conference on Computer Supported Cooperative Work in Design).

Type	Venue	Papers
	IEEE Transactions on Visualization and Computer Graphics	[21, 52, 48, 24,
	(TVCG)	41, 31, 18, 50]
	Computer Graphics Forum	[55, 42, 40, 47]
	IEEE Pacific Visualization Conference (PacificVis)	[34, 44, 56]
	Journal of Visualization	[53, 35]
	IEEE Conference on Visual Analytics Science and Technology	[37]
VIS	IEEE Visualization & Visual Analytics (VIS)	[43]
VID	International Conference Information Visualization (IV)	[36]
	International Conference on Graphics, Patterns and Images SIB- GRAPI	[29]
	International Joint Conference on Computer Vision, Imaging and	[40]
	Computer Graphics Theory and Applications (VISIGRAPP)	[49]
	IEEE Transactions on Intelligent Transportation Systems	[26, 20, 38]
	Journal of Transport Geography	[32, 30, 33]
GIS	ACM International Conference on Advances in geographic Informa- tion Systems (SIGSPATIAL)	[51, 39]
	International Journal of Geographical Information Science	[45]
	Cartography and Geographic Information Science (Taylor & Francis)	[46]
	ACM Transactions on Intelligent Systems and Technology	[23]
CS	MDPI Journal Algorithms	[22]
05	Human-centric Computing and Information Sciences Journal	[54]
	IEEE International Conference on Computer Supported Cooperative	[25]
MULTI	Work in Design (CSCWD)	[20]
	Advances in Cross-Cultural Decision Making	[17]
	Spatial Cognition & Computation (Taylor & Francis)	[27]

Table 7: Classification o	f papers a	according to p	publication	venue (VI	S: Visualization,	GIS:
Geographic Information	Science,	CS: Compu	ter Science,	MULTI:	multi-disciplinar	y.

In VIS and Computer Science-related venues, the motivations are typically centered on the challenges of extracting mobility patterns from heterogeneous data and the design of novel visualization approaches. The latter is also a motivation in the GIS community, which is also often interested in knowledge extraction. Regardless of the venue, the papers often cover mobility-related issues such as urban dynamics, activity patterns, population distribution, crowd behavior, and commuting patterns. VIS and other Computer Science communities often exploit huge datasets extracted from sources such as mobile phone records, social media, POI data, and CDRs. Moreover, the combination of multiple data sources is more prevalent in the VIS and Computer Science communities. In contrast, GIS studies predominantly focus on survey data. The most outstanding difference concerns the user-based evaluations, which seem to be more prevalent within the VIS community (18 out of 22 assessments involving users) compared to the GIS community (only 3 out of 10 assessments with users).

#### 8. Discussion and Research Agenda

In this paper, we surveyed 38 papers covering the visual analysis of individual mobility data. We focused on understanding how the thematic data dimension is integrated into existing visualization solutions. Below, we also highlight potential research perspectives within the domain.

#### 8.1. Theme integration

The six key thematic dimensions covered in the surveyed papers include trip purpose, travel mode, demographics, travel distance, travel duration, and travel speed. While these provide valuable insights into mobility patterns, several critical aspects remain underexplored. For example, spatiotemporal factors like climate change (temperature, air quality, CO2, pollution) can significantly impact movement. Mobility also varies widely, from short commutes to global travel, requiring a broader view of both routine and exceptional movements. Complex patterns, such as those of migrants, often involve interrelated reasons and collaboration among individuals, highlighting the influence of economic conditions and policies. These factors shape mobility by affecting the distribution of services like healthcare and education. Despite the availability of data, these elements are rarely considered in mobility studies. Expanding research to include such correlations could offer a fuller understanding of how policies and environmental factors affect human mobility across different contexts.

#### 8.2. 3D Visualizations

Due to its capacity to represent multiple data dimensions, 3D visualization is inherently an asset for visualizing individual mobility data. In particular, the Space-Time Cube (STC) is a prominent 3D visualization technique in the domain and has been widely employed in the surveyed papers. Although the STC was introduced as early as 1976 [62], research questions about its effectiveness in supporting data exploration remained [63]. The more recent studies by Kveladze [45] and Gonçalvez et al. [64] have shown that combining the STC with 2D representations of data can be beneficial for the analysis and even preferred by the users, who gain multiple perspectives on the data. The main limitation of the STC, and every other 3D representation for that matter, is the perceptual and interactive restrictions imposed by conventional desktop setups [65]. For that, we suggest that *immersive visualizations* and *spatial interaction* can help overcome some of the limitations of the STC.

Immersive Visualization. One option for achieving immersion in the past was through confined spaces surrounded by retro-projected walls, known as CAVE environments. When coupled with head-tracking technology, such setups provided users with an immersive experience [66]. However, more affordable off-the-shelf Head-Mounted Display technologies such as the Meta Quest have exhibited significant traction in the field. This is primarily due to the rapid technical advancements in the industry, leading to increased affordability and ease of integration into the work environments of data analysts [67]. Interactive STCs have been applied to visualize movement trajectories in various domains, including vessel navigation [68], running sports [69], historic events [69], and urban mobility [70, 71, 72, 73]. In individual mobility visualization, the closest approach in the literature involves using an immersive STC to explore simulated trajectories [74]. This immersive method has been shown to improve usability and reduce mental workload. However, it lacks integration of the thematic dimension. Future research should focus on leveraging the benefits of immersive visualization for individual mobility data exploration, where it could significantly enhance analysis.

Spatial Interaction. Spatial interaction facilitated by mobile controllers has the potential to bridge the gap between the physical and virtual information realms [75]. This interaction paradigm involves mapping real-world movements in directions such as forward/backward, up/down, and left/right to manipulate the virtual camera, enabling users to navigate and explore virtual environments more naturally [76, 77]. Although spatial interaction is not widely applied in the surveyed papers, it could enhance user engagement with 3D visualizations. With the widespread availability of smartphones, which can serve as intuitive physical interaction devices, this approach could also become more accessible and impactful.

#### 8.3. Geo-collaborative and Immersive Visualization

As data volumes and complexity grow (in terms of uncertainty, more ambiguous definitions, and larger scopes [78]), tackling analytical problems becomes more challenging and might require the collaboration of multiple experts. The concept of geo-collaborative visualization, where data and models utilized by participants are closely tied to geographical locations [79]) emerges as a suitable approach to enact the collaboration of multiple experts. Collaboration can happen in physical spaces, e.g., in front of wall-sized visualizations [80], or virtually, through video-conference tools. In this context, and building on previous research in this area [78], immersive VR technologies offer new possibilities for collaborative visualization. Although immersive collaborative visualization has been explored in other fields, its use in individual mobility analysis is still underdeveloped, presenting an opportunity for future research.

#### 8.4. Understanding Users Reasoning through Analytical Provenance

Analytical provenance research is dedicated to understanding users' reasoning processes by examining their interactions with visualizations [81]. The surveyed visualization tools often support individual mobility data analysis through various indicators that describe mobility across dimensions like space, time, and theme. Users employ multiple interactive visualization techniques to explore this data, and the specific path of exploration-comprising queries, visualization choices, and interactions-can vary depending on the user's analytical task or profile [45]. Thus, gaining insight into how domain experts utilize a visualization system to answer domain-specific questions, such as through which mobility measures, spatiotemporal granularity levels, and thematic attributes they select and in what sequence, can be valuable for enhancing user experience and effectiveness, leading to the development of tailored exploratory workflows that align well with different tasks and user needs. However, the surveyed papers barely tapped into the potential of analytical provenance. Although Menin et al. [29] records some information, such as when and where a specific visualization was opened, closed, or modified, no provenance studies based on this data have been reported.

### 8.5. Data Management through Knowledge Graphs

The heterogeneous nature of individual mobility data poses a significant challenge for the field. Our analysis revealed that most studies combine data from multiple sources to reconstruct individual movements and enrich them with thematic information. Each paper introduces a distinct data collection and combination process based on (i) the type of data source and (ii) the analytical objectives of the proposed visualization system. In this context, Resource Description Framework (RDF) knowledge graphs (KGs) could be beneficial for data management, through the use of innovative methods for structuring, publishing, discovering, and integrating data. The strength of the RDF lies in its inherent interoperability and capacity to link resources and properties across the web, enabling access to additional information [82]. We observe a growing use of KG-based representations in the domain, such as modeling territorial changes [83], urban mobility events [84], and urban mobility data [85]. Future research should explore KG representations to enhance data transparency and improve analyses of individual mobility data

#### 8.6. Visualization of Spatial Changes underlying Individual Mobility

All surveyed papers concentrate on individual mobility over short time frames, such as days or weeks. While this approach addresses many domainspecific questions, there is a notable absence of longitudinal studies examining mobility behavior over extended periods, like years. This gap likely stems from changes in territorial nomenclatures and geographical boundaries, which hinder the comparability of mobility data over time. Without recalculating past data according to current geographic areas—a complex process that obscures territorial changes—statistical series experience breaks, leading to potential misinterpretations or biases in the statistics if not properly documented [83]. We have yet to encounter visualization solutions that simultaneously convey changes in both mobility and spatial context. Given the importance of understanding historical mobility behavior to support more sustainable transportation modes, we believe this topic warrants further exploration in the future.

#### 9. Conclusion

In this paper, we presented a comprehensive survey of the literature focusing on the visualization of individual mobility data, i.e., data that describes the trajectories of individuals within urban environments. We analyzed 38 papers published since 2010 in GIS and VIS venues. While the literature has extensively covered the spatiotemporal visualization of movement data, relatively little attention has been devoted to the representation of thematic information describing individuals' motivations for mobility.

Our findings indicate that while thematic attributes are frequently included in visualization solutions, they are often represented only partially. We examined the visualizations based on three domain-specific questions for presence dynamics, travel flows, and individual trajectories. Only three solutions effectively assist mobility experts in addressing all three questions. Furthermore, consistent with prior research [61, 86], we observed a deficiency in formal experiments to assess the effectiveness and value of the proposed visualizations for practical applications. Expert-based interviews were conducted in more than half of the papers; however, they typically involved a limited number of participants, highlighting a persistent challenge in the field. Notably, publications in VIS venues exhibited a higher tendency to incorporate evaluations than those in GIS venues.

Based on our analysis of the existing literature, we identified several research opportunities. These include the need for analytical provenance studies to gain insights into user reasoning during data exploration, the development of novel visualizations capable of effectively integrating spatial changes, the design of collaborative setups and virtual environments to support visual analytics, the exploration of 3D visualization using immersive technologies and spatial interaction, and the need of interoperable technologies such as RDF KGs to facilitate data integration from multiple sources.

#### References

- H. Barbosa, M. Barthelemy, G. Ghoshal, C. R. James, M. Lenormand, T. Louail, R. Menezes, J. J. Ramasco, F. Simini, M. Tomasini, Human mobility: Models and applications, Physics Reports 734 (2018) 1-74. URL: https://www.sciencedirect.com/science/article/pii/S037015731830022X. doi:https://doi.org/10.1016/j.physrep.2018.01.001, human mobility: Models and applications.
- [2] J. Novák, L. Sýkora, A City in Motion: Time-Space Activity and Mobility Patterns of Suburban Inhabitants and the Structuration of the Spatial Organization of the Prague Metropolitan Area, Geografiska Annaler: Series B, Human Geography 89 (2007) 147– 168. URL: https://onlinelibrary.wiley.com/doi/abs/10.1111/ j.1468-0467.2007.00245.x. doi:10.1111/j.1468-0467.2007.00245. x, \_eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1111/j.1468-0467.2007.00245.x.
- [3] A. Montanari, Human mobility, global change and local development., Revue belge de géographie (Belgeo) (2005) 7–18.
- [4] H. Yu, S.-L. Shaw, Representing and visualizing travel diary data: A spatio-temporal gis approach, in: 2004 ESRI International User Conference, 2004, pp. 1–13.

- [5] M. G. McNally, C. R. Rindt, The activity-based approach, in: Handbook of Transport Modelling: 2nd Edition, Emerald Group Publishing Limited, 2007, pp. 55–73.
- [6] G. Andrienko, N. Andrienko, P. Bak, D. Keim, S. Kisilevich, S. Wrobe, A conceptual framework and taxonomy of techniques for analyzing movement, Journal of Visual Languages & Computing 22 (2011) 213 – 232. Special Issue on Visual Analytics and Visual Semantics.
- [7] D. J. Peuquet, It's about time: A conceptual framework for the representation of temporal dynamics in geographic information, Annals of the Association of American Geographers 84 (1994) 441–461.
- [8] J. L. Mennis, D. J. Peuquet, L. Qian, A conceptual framework for incorporating cognitive principles into geographical database representation, International Journal of Geographical Information Science 14 (2000) 501–520.
- W. Chen, F. Guo, F.-Y. Wang, A Survey of Traffic Data Visualization, IEEE Transactions on Intelligent Transportation Systems 16 (2015) 2970-2984. URL: http://ieeexplore.ieee.org/document/7120975/. doi:10.1109/TITS.2015.2436897.
- [10] G. Andrienko, Visual Analytics for Transportation: State of the Art and Further Research Directions, IEEE Transactions on Intelligent Transportation Systems (2015). doi:10.1109/TITS.2017.2683539, iSBN: 1524-9050.
- [11] T. Sobral, T. Galvão, J. Borges, Visualization of Urban Mobility Data from Intelligent Transportation Systems, Sensors 19 (2019) 332. URL: https://www.mdpi.com/1424-8220/19/2/332. doi:10.3390/s19020332, number: 2 Publisher: Multidisciplinary Digital Publishing Institute.
- [12] Y. Zheng, W. Wu, Y. Chen, H. Qu, L. M. Ni, Visual Analytics in Urban Computing : An Overview, IEEE Transactions on Big Data 2 (2016) 276–296.
- [13] Z. Deng, D. Weng, S. Liu, Y. Tian, M. Xu, Y. Wu, A survey of urban visual analytics: Advances and future directions, Computa-

tional Visual Media 9 (2023) 3–39. URL: https://doi.org/10.1007/ s41095-022-0275-7. doi:10.1007/s41095-022-0275-7.

- [14] Y. Guo, S. Guo, Z. Jin, S. Kaul, D. Gotz, N. Cao, Survey on visual analysis of event sequence data, IEEE Transactions on Visualization and Computer Graphics 28 (2022) 5091–5112. doi:10.1109/TVCG.2021. 3100413.
- [15] A. Menin, eSTIMe : un environnement de visualisation pour l'analyse multi-points de vue des mobilités quotidiennes, Ph.D. thesis, Université Grenoble Alpes, 2020. URL: http://www.theses.fr/2020GRALS010, thèse de doctorat dirigée par Davoine, Paule-Annick et Nedel, Luciana Ingénierie de la cognition, de l'interaction, de l'apprentissage et de la création Université Grenoble Alpes 2020.
- [16] N. Andrienko, G. Andrienko, Visual analytics of movement: An overview of methods, tools and procedures, Information Visualization 12 (2013) 3–24. doi:10.1177/1473871612457601.
- [17] C. Qian, P. Kats, S. Malinchik, M. Hoffman, B. Kettler, C. Kontokosta, S. Sobolevsky, Geo-tagged social media data as a proxy for urban mobility, in: Advances in Cross-Cultural Decision Making: Proceedings of the AHFE 2017 International Conference on Cross-Cultural Decision Making, July 17-21, 2017, The Westin Bonaventure Hotel, Los Angeles, California, USA 8, Springer, 2018, pp. 29–40.
- [18] L. Shi, C. Huang, M. Liu, J. Yan, T. Jiang, Z. Tan, Y. Hu, W. Chen, X. Zhang, Urbanmotion: Visual analysis of metropolitan-scale sparse trajectories, IEEE Transactions on Visualization and Computer Graphics 27 (2020) 3881–3899.
- [19] J. Sammons, Mobile device forensics, in: The Basics of Digital Forensics, 2nd ed., Syngress, 2015, pp. 145 – 161.
- [20] H. Senaratne, M. Mueller, M. Behrisch, F. Lalanne, J. Bustos-Jiménez, J. Schneidewind, D. Keim, T. Schreck, Urban mobility analysis with mobile network data: A visual analytics approach, IEEE Transactions on Intelligent Transportation Systems 19 (2017) 1537–1546.
- [21] S. Chen, X. Yuan, Z. Wang, C. Guo, J. Liang, Z. Wang, X. Zhang, J. Zhang, Interactive visual discovering of movement patterns from

sparsely sampled geo-tagged social media data, IEEE transactions on visualization and computer graphics 22 (2015) 270–279.

- [22] I. Pérez-Messina, E. Graells-Garrido, M. J. Lobo, C. Hurter, Modalflow: cross-origin flow data visualization for urban mobility, Algorithms 13 (2020) 298.
- [23] W. Chen, J. Xia, X. Wang, Y. Wang, J. Chen, L. Chang, Relationlines: Visual reasoning of egocentric relations from heterogeneous urban data, ACM Transactions on Intelligent Systems and Technology (TIST) 10 (2018) 1–21.
- [24] W. Chen, Z. Huang, F. Wu, M. Zhu, H. Guan, R. Maciejewski, Vaud: A visual analysis approach for exploring spatio-temporal urban data, IEEE transactions on visualization and computer graphics 24 (2017) 2636–2648.
- [25] X. Zhang, Q. Wang, Peoplevis: A visual analysis system for mining travel behavior, in: 2017 IEEE 21st International Conference on Computer Supported Cooperative Work in Design (CSCWD), IEEE, 2017, pp. 463–468.
- [26] F. Wang, W. Chen, Y. Zhao, T. Gu, S. Gao, H. Bao, Adaptively exploring population mobility patterns in flow visualization, IEEE Transactions on Intelligent Transportation Systems 18 (2017) 2250–2259.
- [27] S. Gao, Spatio-temporal analytics for exploring human mobility patterns and urban dynamics in the mobile age, Spatial Cognition & Computation 15 (2015) 86–114.
- [28] Certu, Les enquêtes déplacements "standard certu", https: //www.cnis.fr/wp-content/uploads/2017/12/DPR\_2013\_2e\_ reunion\_COM\_territoires\_enquetes\_deplacements\_certu.pdf, 2013. Accessed on 29th July 2020.
- [29] A. Menin, S. Chardonnel, P.-A. Davoine, L. Nedel, Estime: Towards an all-in-one geovisualization environment for daily mobility analysis, in: 2019 32nd SIBGRAPI Conference on Graphics, Patterns and Images (SIBGRAPI), IEEE, 2019, pp. 39–46.

- [30] G. Le Roux, J. Vallée, H. Commenges, Social segregation around the clock in the paris region (france), Journal of Transport Geography 59 (2017) 134–145.
- [31] Z. Huang, Y. Zhao, W. Chen, S. Gao, K. Yu, W. Xu, M. Tang, M. Zhu, M. Xu, A natural-language-based visual query approach of uncertain human trajectories, IEEE Transactions on Visualization and Computer Graphics 26 (2019) 1256–1266.
- [32] J. Chen, S.-L. Shaw, H. Yu, F. Lu, Y. Chai, Q. Jia, Exploratory data analysis of activity diary data: a space-time gis approach, Journal of Transport Geography 19 (2011) 394–404.
- [33] K. Liu, Y. Murayama, T. Ichinose, A multi-view of the daily urban rhythms of human mobility in the tokyo metropolitan area, Journal of Transport Geography 91 (2021) 102985.
- [34] S. Gupta, M. Dumas, M. J. McGuffin, T. Kapler, Movementslicer: Better gantt charts for visualizing behaviors and meetings in movement data, in: 2016 IEEE Pacific Visualization Symposium (PacificVis), IEEE, 2016, pp. 168–175.
- [35] Z. Deng, D. Weng, Y. Wu, You are experienced: Interactive tour planning with crowdsourcing tour data from web, Journal of Visualization 26 (2023) 385–401.
- [36] K. Vrotsou, C. Forsell, M. Cooper, 2d and 3d representations for feature recognition in time geographical diary data, Information Visualization 9 (2010) 263–276.
- [37] L. Yu, W. Wu, X. Li, G. Li, W. S. Ng, S.-K. Ng, Z. Huang, A. Arunan, H. M. Watt, iviztrans: Interactive visual learning for home and work place detection from massive public transportation data, in: 2015 IEEE Conference on Visual Analytics Science and Technology (VAST), IEEE, 2015, pp. 49–56.
- [38] W. Zeng, C.-W. Fu, S. M. Arisona, S. Schubiger, R. Burkhard, K.-L. Ma, Visualizing the relationship between human mobility and points of interest, IEEE Transactions on Intelligent Transportation Systems 18 (2017) 2271–2284.

- [39] Y. Santos, R. Giuliani, T. Portela, C. Renso, J. Carvalho, Mat-ca: a tool for multiple aspect trajectory clustering analysis, in: Proceedings of the 1st ACM SIGSPATIAL International Workshop on Methods for Enriched Mobility Data: Emerging issues and Ethical perspectives 2023, 2023, pp. 40–43.
- [40] R. Krueger, Q. Han, N. Ivanov, S. Mahtal, D. Thom, H. Pfister, T. Ertl, Bird's-eye-large-scale visual analytics of city dynamics using social location data, Computer Graphics Forum 38 (2019) 595–607.
- [41] C. Yang, Z. Zhang, Z. Fan, R. Jiang, Q. Chen, X. Song, R. Shibasaki, Epimob: Interactive visual analytics of citywide human mobility restrictions for epidemic control, IEEE Transactions on Visualization and Computer Graphics (2022).
- [42] W. Zeng, C.-W. Fu, S. Müller Arisona, A. Erath, H. Qu, Visualizing waypoints-constrained origin-destination patterns for massive transportation data, Computer Graphics Forum 35 (2016) 95–107.
- [43] F. Gortana, S. Kaim, M. von Lupin, T. Nagel, Isoscope-visualizing temporal mobility variance with isochrone maps, Poster Abstracts of IEEE VIS 2014 (2014).
- [44] W. Wu, Y. Zheng, N. Cao, H. Zeng, B. Ni, H. Qu, L. M. Ni, Mobiseg: Interactive region segmentation using heterogeneous mobility data, in: 2017 IEEE Pacific visualization symposium (PacificVis), IEEE, 2017, pp. 91–100.
- [45] I. Kveladze, M.-J. Kraak, C. P. Van Elzakker, The space-time cube as part of a geovisual analytics environment to support the understanding of movement data, International Journal of Geographical Information Science 29 (2015) 2001–2016.
- [46] L. Conrow, C. Fu, H. Huang, N. Andrienko, G. Andrienko, R. Weibel, A conceptual framework for developing dashboards for big mobility data, Cartography and geographic information science 50 (2023) 495–514.
- [47] C. Lv, H. Zhang, Y. Lin, J. Dong, L. Tian, Routevis: Quantitative visual analytics of various factors to understand route choice preferences, Computer Graphics Forum 43 (2024) e15091.

- [48] W. Wu, J. Xu, H. Zeng, Y. Zheng, H. Qu, B. Ni, M. Yuan, L. M. Ni, Telcovis: Visual exploration of co-occurrence in urban human mobility based on telco data, IEEE transactions on visualization and computer graphics 22 (2015) 935–944.
- [49] P. Silva, C. Maças, J. Correia, P. Machado, E. Polisciuc, Antenna: A tool for visual analysis of urban mobility based on cell phone data., in: VISIGRAPP (3: IVAPP), 2022, pp. 88–100.
- [50] Z. Zhou, L. Meng, C. Tang, Y. Zhao, Z. Guo, M. Hu, W. Chen, Visual abstraction of large scale geospatial origin-destination movement data, IEEE transactions on visualization and computer graphics 25 (2018) 43–53.
- [51] X. Liu, Q. Huang, Z. Li, M. Wu, The impact of mtup to explore online trajectories for human mobility studies, in: Proceedings of the 1st ACM SIGSPATIAL Workshop on Prediction of Human Mobility, 2017, pp. 1– 9.
- [52] G. Di Lorenzo, M. Sbodio, F. Calabrese, M. Berlingerio, F. Pinelli, R. Nair, Allaboard: Visual exploration of cellphone mobility data to optimise public transport, IEEE transactions on visualization and computer graphics 22 (2015) 1036–1050.
- [53] F. Wu, M. Zhu, Q. Wang, X. Zhao, W. Chen, R. Maciejewski, Spatialtemporal visualization of city-wide crowd movement, Journal of Visualization 20 (2017) 183–194.
- [54] L. Shi, Z. Guo, T. Jiang, R. Fang, Y. Chen, Y. Zhao, X. Zhang, M. Dai, Visual analysis of steady-state human mobility in cities, HUMAN-CENTRIC COMPUTING AND INFORMATION SCIENCES 11 (2021).
- [55] W. Zeng, C.-W. Fu, S. M. Arisona, H. Qu, Visualizing interchange patterns in massive movement data, Computer Graphics Forum 32 (2013) 271–280.
- [56] D.-B. Chen, C.-H. Lai, Y.-H. Lien, Y.-H. Lin, Y.-S. Wang, K.-L. Ma, Representing multivariate data by optimal colors to uncover events of interest in time series data, in: 2020 IEEE Pacific Visualization Symposium (PacificVis), IEEE, 2020, pp. 156–165.

- [57] R. Borgo, J. Kehrer, D. H. S. Chung, E. Maguire, R. S. Laramee, H. Hauser, M. Ward, M. Chen, Glyph-based Visualization: Foundations, Design Guidelines, Techniques and Applications, Eurographics State of the Art Reports (2013) 39-63. URL: http://www.cg.tuwien.ac.at/ research/publications/2013/borgo-2013-gly/, iSBN: 1558608192.
- [58] T. Hägerstrand, What about people in regional science?, Regional Science Association 24 (1970) 6–21.
- [59] T. Munzner, Visualization analysis and design, CRC press, 2014.
- [60] W. Trochim, J. Donnelly, K. Arora, Research methods: The essential knowledge base, 2nd editio. ed, 2015.
- [61] G. Ellis, A. Dix, An explorative analysis of user evaluation studies in information visualisation, in: Proceedings of the 2006 AVI workshop on BEyond time and errors: novel evaluation methods for information visualization, 2006, pp. 1–7.
- [62] T. Hägerstraand, What about people in regional science?, Papers in Regional Science 24 (1970) 7–24.
- [63] M.-J. Kraak, The space-time cube revisited from a geovisualization perspective, in: Proceedings of the 21st International Cartographic Conference (ICC), International Cartographic Association, 2003, pp. 1988– 1996.
- [64] T. Gonçalves, A. P. Afonso, B. Martins, Why not both?-combining 2d maps and 3d space-time cubes for human trajectory data visualization, in: Proceedings of the 30th International BCS Human Computer Interaction Conference 30, 2016, pp. 1–10.
- [65] C. Ware, Information visualization: perception for design, Morgan Kaufmann, 2019.
- [66] K. Reda, A. Febretti, A. Knoll, J. Aurisano, J. Leigh, A. Johnson, M. E. Papka, M. Hereld, Visualizing large, heterogeneous data in hybrid-reality environments, IEEE Computer Graphics and Applications 33 (2013) 38–48. doi:10.1109/MCG.2013.37.

- [67] D. Zielasko, B. Weyers, M. Bellgardt, S. Pick, A. Meibner, T. Vierjahn, T. W. Kuhlen, Remain seated: Towards fully-immersive desktop VR, in: IEEE 3rd Workshop on Everyday Virtual Reality (WEVR), IEEE, 2017, pp. 1–6. doi:10.1109/WEVR.2017.7957707.
- [68] N. Willems, H. Van De Wetering, J. J. Van Wijk, Evaluation of the visibility of vessel movement features in trajectory visualizations, Computer Graphics Forum 30 (2011) 801–810. doi:10.1111/j.1467-8659. 2011.01929.x.
- [69] M.-J. Kraak, Geovisualization and time new opportunities for the space-time cube, Geographic visualization: Concepts, tools and applications (2008) 293–306. doi:10.1002/9780470987643.ch15.
- [70] F. Amini, S. Rufiange, Z. Hossain, Q. Ventura, P. Irani, M. J. McGuffin, The impact of interactivity on comprehending 2D and 3D visualizations of movement data, IEEE Transactions on Visualization and Computer Graphics 21 (2015) 122–135. doi:10.1109/TVCG.2014.2329308.
- [71] I. Kveladze, M. Kraak, C. van Elzakker, Researching the usability of a geovisual analytics environment for the exploration and analysis of different datasets, in: The 28th International Cartographic Conference, 2-7 July 2017, Washington DC, USA, 2017, pp. 2–7.
- [72] T. Gonçalves, A. P. Afonso, B. Martins, Cartographic visualization of human trajectory data: Overview and analysis, Journal of Location Based Services 9 (2015) 138–166. doi:10.1080/17489725.2015. 1074736.
- [73] J. Wagner, C. T. Silva, W. Stuerzlinger, L. Nedel, Reimagining taxivis through an immersive space-time cube metaphor and reflecting on potential benefits of immersive analytics for urban data exploration, in: 2024 IEEE Conference Virtual Reality and 3D User Interfaces (VR), 2024, pp. 827–838. doi:10.1109/VR58804.2024.00102.
- [74] J. A. Wagner Filho, W. Stuerzlinger, L. Nedel, Evaluating an immersive space-time cube geovisualization for intuitive trajectory data exploration, IEEE transactions on visualization and computer graphics 26 (2019) 514–524.

- [75] P. Fröhlich, R. Simon, L. Baillie, J. Roberts, R. Murray-Smith, Mobile spatial interaction, in: CHI'07 Extended Abstracts on Human Factors in Computing Systems, 2007, pp. 2841–2844.
- [76] L. Besançon, P. Issartel, M. Ammi, T. Isenberg, Hybrid tactile/tangible interaction for 3d data exploration, IEEE transactions on visualization and computer graphics 23 (2017) 881–890.
- [77] M. Pahud, E. Ofek, N. H. Riche, C. Hurter, J. Grubert, Mobiles as Portals for Interacting with Virtual Data Visualizations, arXiv preprint arXiv:1804.03211 (2018).
- [78] P. Isenberg, N. Elmqvist, J. Scholtz, D. Cernea, K.-L. Ma, H. Hagen, Collaborative visualization: Definition, challenges, and research agenda, Information Visualization 10 (2011) 310–326.
- [79] N. Baloian, J. Frez, J. A. Pino, G. Zurita, Supporting Collaborative Decision Making in Geo-Collaboration Scenarios, in: N. Baloian, Y. Zorian, P. Taslakian, S. Shoukouryan (Eds.), Collaboration and Technology, Lecture Notes in Computer Science, Springer International Publishing, Cham, 2015, pp. 63–71. doi:10.1007/978-3-319-22747-4\_5.
- [80] R. Langner, U. Kister, R. Dachselt, Multiple coordinated views at large displays for multiple users: Empirical findings on user behavior, movements, and distances, IEEE transactions on visualization and computer graphics 25 (2018) 608–618.
- [81] C. North, R. Chang, A. Endert, W. Dou, R. May, B. Pike, G. Fink, Analytic provenance: process+ interaction+ insight, in: CHI'11 Extended Abstracts on Human Factors in Computing Systems, 2011, pp. 33–36.
- [82] F. Gandon, A survey of the first 20 years of research on semantic Web and linked data, Ingénierie des systèmes d'information 23 (2018) 11-38. URL: https://isi.revuesonline.com/article.jsp?articleId= 39845. doi:10.3166/isi.23.3-4.11-38.
- [83] C. Bernard, Immersing evolving geographic divisions in the semantic Web, Theses, Université Grenoble Alpes, 2019. URL: https://theses. hal.science/tel-02524361.

- [84] T. Sobral, T. Galvão, J. Borges, VUMO: Towards an ontology of urban mobility events for supporting semi-automatic visualization tools, in: 2016 IEEE 19th International Conference on Intelligent Transportation Systems (ITSC), IEEE, Rio de Janeiro, Brazil, 2016, pp. 1700–1705. URL: https://ieeexplore.ieee.org/ document/7795787/. doi:10.1109/ITSC.2016.7795787.
- [85] Q. Zhang, Z. Ma, P. Zhang, E. Jenelius, Mobility knowledge graph: review and its application in public transport, Transportation (2023). URL: https://doi.org/10.1007/s11116-023-10451-8. doi:10.1007/ s11116-023-10451-8.
- [86] K. Andrews, Evaluating information visualisations, in: Proceedings of the 2006 AVI workshop on BEyond time and errors: novel evaluation methods for information visualization, 2006, pp. 1–5.